Week 34  

SM18 Operation SCHEDULE Meeting (Tuesdays)

 Bench status  and tests for this week : 

	TBA1
	MBBL3192
	ICS 5 Traveller Last
	300.8

	TBA2
	MBBL1107
	Prep 5 Pumping + Cool Down
	1.9

	TBB1
	SSS041
	ICS 5 Traveller Last
	300.3

	TBB2
	MBBL1121
	PT 12 Warm Up
	152.9

	TBC1
	MBAL3079
	PT 1 IAP @ cold
	1.91

	TBC2
	MBBL3198
	Prep 5 Pumping + Cool Down
	301.3

	TBD1
	SSS007
	MM3 Full Load Line + Joints
	1.9

	TBD2
	SSS012
	Stretched Wire Warm SW24 survey
	288.7

	TBE1
	MBBL2075
	PT 12 Warm Up
	276.3

	TBE2
	MBBR1019
	Prep 5 Pumping + Cool Down
	302.1

	TBF1
	MBBL3136
	CR 1 Leak Test
	223.4

	TBF2
	MBBL1131
	Prep 5 Pumping + Cool Down
	104.8


Test Coordinator this week:  M.Buzio;  MTM  Support. Maryline:  (16-3602) . 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

1. Problem with CFU: cooling down to 80 K was slow Sunday. One parameter changed in the process.
· SSS
1. SSS: Survey called Sunday morning : not responding all the day. Finally could be reached Sunday night. One CERN number now 165010.
2. Survey on SSS41: delay of one day to perform the survey in aperture 2.

· AOB
1. DAQ for Training quench : Pre-trigger time increased to 154 s: saving DAQ file will take longer.
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Wednesday_11_M


	TBC2
	IAP @ warm : interlock present in pot-aim cards
	ICS
	
	Guy informed. He asked ICS to check the connection. A wire was broken on CFB side. Repaired by ICS

	Thursday_12_A
	TBB1
	PLC-Art communication problem.
	AB/CO
	
	Put off and on. Then OK

	Thursday _12_N
	TBE1
	MM1: VME error. Remains after resetting the crates.
	AB/CO
	
	Guy informed.

	Friday_13_M
	TBB2
	Tcoil measurement: wrong value of T on slots 4 and 8 aperture 2
	MTM
	
	Guy changed the faulty cards.

	Friday_13_M
	TBB1
	SSS41: SPA: bench configured in dipole for the inductance
	MTM
	
	Giorgio changed for SSS.

	Friday_13_M
	TBB2
	Permanent trigger on LF:
	MTM
	
	Giorgio found that card Crate1 slot 16 was faulty.

	Friday_13_N

Saturday_14_M
	TBE1
	Saving of training quench files is longer than usual.

2074: AQA warning ; 

Tau jumps ~9 V

VT exceeds 100 V (D2_L~109 V)
	MTM
	
	Reason: pre-trigger time increased from 9s to 157 s with respect to previous config file. It will take now longer to receive the HF files.

Giorgio informed.He asked to check QH discharge and MIITS.

	Friday_13_N
	TBD1
	All 4 heaters should fire after trigger.
	MTM
	
	Giorgio consulted.

	Friday_13_N
	TBD1
	LHC cycle not OK.

DAQ trigger not received

Motor touching the end stops

Motor 2 touching the end stops

Over-range in many sectors
	MTM
	
	Wrong setting was selected in MMP (dipole instead of quadrupole). GHH restart the measurements.

	Saturday_14_A
	TBE2
	IAP @ warm on 1019 magnet : Temperature sensor reading not OK (173 K instead of 300 K).
	
	
	Giorgio consulted. He checked the resistance: not OK 113 ohms instead of 60 ohms. NC opened.

	Saturday_15_A
	TBD1
	600 A power supply cannot be controlled automatically. Piloted only through a terminals

Then coil A in sector 0001 in ap2 not working :
	AB/PO
	
	Problem in gateway.

Changed in E-coil.








